Tronox Daily Report



Project: **BMI003 - Tronox**Date: **30-Aug-10**

Personnel: D. Gordon

Temperature: Hi:91 Lo:65
Weather: Mostly clear
Wind: Light breeze
Dust: None

Significant Issues: None.		

Observations:

New Activity: Using a John Deer skid steer to start cleaning the floor of the basement.

Continued Activity Day shift: Extending Process Road to intersect 4th street. Gate installation work.

Continued Activity Night shift: ACM remediation in the Unit 1 basement area. The excavator continued digging and stockpiling the material to the east for load out. Good water support from the laborer working the hose. Loader operator placed the material in the lined OTR trucks taking care not to disturb the plastic. Two laborers in proper PPE place a sheet of plastic down the center of the load they bring in the side plastic. Sealing all edged with spray adhesive and tape. Air monitoring stations at the 4 corners of the work area.

Dust Control/Stockpile Management: Hard to manage; stop-work was called.

Track out: Minimal.

Other:	
--------	--

Night shift ACM Load count: 58





LVP water truck helping wet the work area. Looking south from observation lift.



Looking east, Skid Steer working the floor.





Good water suport.



Floor at end of day. There will be shovel work to finish the floor.



Project: **BMI003 - Tronox**Date: **30-Aug-10**Personnel: **D. Gordon**

Equipment & Materials Usage:			
Miles	26		
Digital Camera	1		
PID			
Sample Kit			
Bailers			
Level C			
Level B			

Site arrivals/departures:

For D. Gordon: Onsite 19:30 Offsite: 21:30 Onsite: 00:30 Offsite: 04:00

Onsite: Offsite: